6.875J/18.425J Cryptography and Cryptanalysis September 5, 2001

Handout 1c: Sign-up Sheet

Instructor: Anna Lysyanskaya Teaching Assistant: Moses Liskov

Please fill in the following information, and return this sheet to us by the end of class today.

Name:

Email:

(circle one:) Undergraduate student / M. Eng student / PhD student

Course:

Likelihood you will remain in this course:

Background (circle all that apply):

Basic Probability (6.042) Algorithms (6.046)

Complexity (6.045/6.840) Number Theory or Abstract Algebra (18.781/18.701)

1c-1